Author’s response to reviews

Title: Treatment of Urinary Calculi After Yang-Monti Ileal Ureter Reconstruction: A Case Report

Authors:

zhenxing wang (76166604@qq.com)
guangheng luo (13809440294@163.com)
ye tian (tianye5055@163.com)
xiushu yang (13985037039@139.com)

Version: 3 Date: 12 Dec 2018

Author’s response to reviews:

Response Letter

Dear Editors,

Thank you very much for your letter and advice. We have revised the manuscript and would like to resubmit it for your consideration. We have addressed the comments raised by the reviewers, and the amendments are highlighted in red in the revised manuscript. Our point-by-point responses to the reviewers’ comments are listed below.

We have completed our change of authorship form, and sent it to BMCSeriesEditorial@biomedcentral.com (the form was in the letter attachments). We hope that this change is acceptable.

This manuscript has been professionally edited and proofread by Kelly Zammit, BVSc, and Angela Morben, DVM, ELS, from Liwen Bianji, Edanz Editing China (www.liwenbianji.cn/ac).

We hope that the revised version of the manuscript is now acceptable for publication in your journal. We look forward to hearing from you soon.
Yours sincerely,

Sun Zhaolin and Luo Guangheng

First and second corresponding authors

E-mail addresses

SZ: 763920818@qq.com
LG: 13809440294@163.com

We would like to express our sincere thanks to the reviewers for the constructive and positive comments.

Responses to the comments from Reviewer 1

Yasser Noureldin (Reviewer 1): Thank you for addressing the required reviews!

Reply: Thank you for your positive comment.

Responses to the comments from Reviewer 3

Eyüp Burak Sancak (Reviewer 3): English grammatical errors are uncorrected and still have some meaningless words (extractor..). Manuscript was checked inelaborately by yours.

Reply: We have revised the English and grammar throughout the manuscript, and the manuscript has been professionally edited by a native English speaker.

In abstract section, the sentences 'ipsilateral lower ureteral calculi and lower ureteral calculi after Yang-Monti ileal ureteroscopy' still was not corrected.

Reply: We have corrected this sentence.(line 7, page2)